A paired-kidney allocation study found superior survival with HLA-DR compatible kidney transplants in the Eurotransplant Senior Program



OPEN

Johan de Fijter^{1,32}, Geertje Dreyer¹, Marko Mallat¹, Klemens Budde², Johann Pratschke³, Jürgen Klempnauer⁴, Martin Zeier⁵, Wolfgang Arns⁶, Christian Hugo⁷, Lars-Christian Rump⁸, Ingeborg Hauser⁹, Peter Schenker¹⁰, Mario Schiffer¹¹, Marc-Oliver Grimm¹², Volker Kliem¹³, Christoph J. Olbricht¹⁴, Przemyslaw Pisarski¹⁵, Bernhard Banas¹⁶, Barbara Suwelack¹⁷, Oliver Hakenberg¹⁸, Gabriela Berlakovich¹⁹, Stefan Schneeberger²⁰, Jacqueline van de Wetering²¹, Stefan Berger²², Frederike Bemelman²³, Dirk Kuypers²⁴, Sebastiaan Heidt²⁵, Axel Rahmel^{26,33}, Frans Claas²⁵, Patrick Peeters²⁷, Rainer Oberbauer²⁸, Uwe Heemann²⁹ and Bernhard K. Krämer³⁰; on behalf of the Eurotransplant Senior DR-compatible Program (ESDP) Study Group³¹

¹Department of Nephrology, Leiden University Medical Center, Leiden, Netherlands; ²Department of Nephrology, Internal Intensive Care Medicine, Campus Charité Mitte, Berlin, Germany; ³Department of Surgery, Campus Charité Mitte and Campus Virchow Klinikum, Charité - Universitätsmedizin Berlin, Berlin, Germany; ⁴Integrated Research and Treatment Centre Transplantation, Hannover Medical School, Hannover, Germany; ⁵Department of Nephrology, Heidelberg University Hospital, Heidelberg, Germany; ⁶Department of Nephrology and Transplantation, Cologne Merheim Medical Center, Cologne, Germany; ⁷Clinic for Internal Medicine III, University Hospital Carl Gustav Carus, Dresden, Germany; ⁸Department of Internal Medicine/Nephrology, Universitätsklinikum Düsseldorf, Düsseldorf, Germany; ⁹Department of Nephrology, Goethe University Hospital Frankfurt, Frankfurt/Main, Germany; ¹⁰Department of Surgery, University Hospital Knappschaftskrankenhaus Bochum, Ruhr-University Bochum, Bochum, Germany; ¹¹Department of Nephrology, Erlangen University Hospital, Erlangen, Germany; ¹²Department of Urology, Jena University Hospital, Jena, Germany; ¹³Department of Internal Medicine and Nephrology, Kidney Transplant Center, Nephrological Center of Lower Saxony, Klinikum Hann, Münden, Germany; ¹⁴Transplant Center, Klinikum Stuttgart, Stuttgart, Germany; ¹⁵Department of Surgery, Section of Transplant Surgery, Medical Center— University of Freiburg, Freiburg, Germany; ¹⁶Department of Nephrology, University Hospital Regensburg, Regensburg, Germany; ¹⁷Department of Internal Medicine, Nephrology and Rheumatology, University Hospital of Münster, Münster, Germany; ¹⁸Department of Urology, Universitätsmedizin, Rostock, Germany; ¹⁹Division of Transplantation, Department of Surgery, Medical University of Vienna, Wien, Austria; ²⁰Department of Visceral, Transplant and Thoracic Surgery, Medical University of Innsbruck, Innsbruck, Austria; ²¹Department of Nephrology and Transplantation, Erasmus University Medical Center, Rotterdam, Netherlands; ²²Department of Nephrology, University Medical Center Groningen, Groningen, Netherlands; ²³Department of Nephrology, Division of Internal Medicine, Amsterdam University Medical Center, Amsterdam, Netherlands; ²⁴Department of Nephrology, University Hospitals Leuven, Leuven, Belgium; ²⁵Eurotransplant Reference Laboratory, Leiden University Medical Center, Leiden, Netherlands; ²⁶Eurotransplant International Foundation, Leiden, Netherlands; ²⁷Department of Nephrology, Ghent University Hospital, Ghent, Belgium; ²⁸Department of Nephrology, Medical University of Vienna, Vienna, Austria;²⁹Department of Nephrology, Klinikum rechts der Isar, Technische Universität München, Munich, Germany; and ³⁰V-th Department of Medicine (Nephrology), University Medical Center Mannheim/University of Heidelberg, Mannheim, Germany

The Eurotransplant Senior Program (ESP) has expedited the chance for elderly patients with kidney failure to receive a timely transplant. This current study evaluated survival parameters of kidneys donated after brain death with or without matching for HLA-DR antigens. This cohort study evaluated the period within ESP with paired allocation of 675 kidneys from donors 65 years and older to transplant

³¹Members of the Eurotransplant Senior DR-compatible Program (ESDP) Study Group are listed in the Appendix.

³²Current affiliation: Antwerp University Hospital, Antwerpen, Belgium.

³³Current address: Deutsche Stiftung Organtransplantation, DSO, Deutschherrnufer 52, D-60594 Frankfurt, Germany.

Received 18 August 2022; revised 27 March 2023; accepted 4 May 2023; published online 19 June 2023

candidates 65 years and older, the first kidney to 341 patients within the Eurotransplant Senior DR-compatible Program and 334 contralateral kidneys without (ESP) HLA-DR antigen matching. We used Kaplan-Meier estimates and competing risk analysis to assess all cause mortality and kidney graft failure, respectively. The log-rank test and Cox proportional hazards regression were used for comparisons. Within ESP, matching for HLA-DR antigens was associated with a significantly lower five-year risk of mortality (hazard ratio 0.71; 95% confidence interval 0.53-0.95) and significantly lower cause-specific hazards for kidney graft failure and return to dialysis at one year (0.55; 0.35-0.87) and five years (0.73; 0.53-0.99) post-transplant. Allocation based on HLA-DR matching resulted in longer cold ischemia (mean difference 1.00 hours; 95% confidence interval: 0.32-1.68) and kidney offers with a significantly shorter median dialysis vintage of 2.4 versus 4.1 yrs. in ESP without matching. Thus, our allocation based on HLA-DR

Correspondence: Johan (Hans) de Fijter, Department of Nephrology and Hypertension, Antwerp University Hospital, Drie Eikenstraat 655, 2650 Edegem, Antwerpen, Belgium. E-mail: Hans.deFijter@uza.be

matching improved five-year patient and kidney allograft survival. Hence, our paired allocation study suggests a superior outcome of HLA-DR matching in the context of old-for-old kidney transplantation.

Kidney International (2023) **104,** 552–561; https://doi.org/10.1016/ j.kint.2023.05.025

KEYWORDS: allocation; Eurotransplant Senior Program; histocompatibility; HLA-DR matching; kidney transplantation; old-for-old allocation

Copyright © 2023, International Society of Nephrology. Published by Elsevier Inc. This is an open access article under the CC BY license (http:// creativecommons.org/licenses/by/4.0/).

Lay Summary

The elderly represent the fastest-growing subpopulation with kidney failure, but only a limited number of such patients receive a kidney transplant in time. Waiting time on dialysis and dialysis vintage are associated with a higher incidence of mortality after a successful kidney transplant. The Eurotransplant Senior Program (ESP) preferentially allocates kidneys from donors aged ≥ 65 years to elderly recipients aged \geq 65 years and successfully expedited the chance for these patients to receive a timely transplant. Elderly patients, however, have a higher risk of kidney allograft rejection as well as infection. "Old-for-old" kidney transplant recipients experience a higher incidence of morbidity and excess mortality. The current allocation study investigated whether matching for human leukocyte antigens (HLAs), in particular HLA-DR, within the ESP program could improve the outcome parameters. We randomly allocated 675 kidneys from donors aged ≥65 years to transplant candidates aged \geq 65 years who were on the ESP waiting list. The paired allocation algorithm assigned the first kidney to a patient without mismatches for HLA-DR antigens (i.e., the longest waiting candidate), and the contralateral kidney was allocated following the current ESP standard, based on waiting time only, without matching for HLA antigens. The results showed that allocation based on HLA-DR matching was associated with a 30% lower chance of mortality 5 years after transplantation and with more functioning kidney transplants. This paired allocation study for the first time documented a superior outcome of HLA-DR matching in the context of "old-for-old" kidney transplantation.

he elderly represent the fastest-growing kidney failure population, with over 50% of the new patients entering renal replacement programs being aged 65 years or older.^{1,2} In selected elderly candidates, transplantation adds life-years, and provides better quality of life, and transplanting a kidney of any quality is cost-effective compared to dialysis.^{3–8} However, only a minority of elderly patients with kidney failure are eventually listed for transplantation.¹ Along with the ongoing shortage of organs available for transplantation, elderly transplant candidates face another critical disadvantage. Not only is waiting time while on dialysis associated with significant mortality incidence, but also, dialysis vintage reduces graft survival and overall life expectancy after transplantation.^{9,10}

Some degree of intuitive age matching, including "old-forold" allocation, has always been part of clinical decisionmaking in most Western countries. Worldwide, the first initiative to expedite the chance for elderly patients to receive a kidney transplant was the Eurotransplant Senior Program (ESP) initiated more than 20 years ago. Preferential allocation of kidneys from donors aged ≥ 65 years to elderly recipients aged ≥ 65 years (old-for-old) significantly increased the chance of receiving a timely transplant.¹¹

A serious threat to older kidney transplant recipients is the excess mortality incidence due to infectious causes, which is the consequence of overimmunosuppression in the light of an already senescent immune system.¹²⁻¹⁴ In addition, recipients of kidneys from older donors experience more acute rejection episodes within every recipient age-category.¹⁵⁻¹⁷ Kidnevs from older donors are more vulnerable to both cellular and humoral rejection mechanisms and carry an increased risk of graft failure.¹⁸⁻²⁰ Although expanded-criteria donor kidneys have been reported to provide, on average, 5 added years of life in general,²¹ additional courses of high-dose corticosteroids and/or T cell-depleting antibodies may substantially increase the incidence of infectious morbidity and mortality in elderly recipients. In fact, although the 5-year analysis of the ESP data has documented acceptable kidney graft survival results, the incidence of infection-related mortality was approximately 30%—that is, quite high.^{11,22}

Current ESP allocation preferentially allocates kidneys from older donors (aged \geq 65 years) without prospective human leukocyte antigen (HLA) matching, to older (aged \geq 65 years) local transplant candidates. Historically, the minimal standard of allocation within Eurotransplant has been at least a 2 DR or 1 B/1 DR match for HLA antigens. To reduce immunogenicity within ESP, the kidney advisory board advised that the ESP allocation algorithm be temporarily changed to include HLA-DR matching. The current study evaluated clinical outcomes of paired kidney allocation with or without matching for HLA class-II antigens within the context of ESP kidney transplantation.

METHODS

Study population and allocation

Between January 1, 2010 and January 4, 2014, a total of 675 recipients, aged \geq 65 years, received a kidney via the paired Eurotransplant Senior DR-compatible Program (ESDP)/ESP kidney allocation algorithm. Only kidneys donated after brain death (DBD) were allocated per donor, with the first kidney being given to the longest waiting patient on the list with a zero HLA-DR mismatch (ESDP). The contralateral kidney was allocated according to the current ESP policy, which is allocation by waiting time only, without consideration of HLA-matching. Matching for HLA-DR was performed at the split antigen level, except for HLA-DR3, for which matching was performed at the broad antigen level. HLA-DR and -DQ typing was performed by using sequence-specific oligonucleotide (SSO) and sequence-specific primer (SSP) DNA-based

techniques. For all the patients and donors included, a typing result for the HLA-DR locus was present, and for \geq 90%, a result for HLA-DQ was present. HLA-DR3 could not be split in HLA-DR17/DR18. In this paired allocation study of donor kidneys with or without matching for HLA-DR antigens, the recipients' characteristics, transplant-specific parameters, and outcome parameters were collected independently. Allocation was on 6 HLA antigens (A, B and DR; not including HLA-DQ or -DP). The allocation script further included identical ABO blood groups and stratification for male or female donor as well as left or right kidney, while keeping cold ischemia time as short as possible but always under 20 hours.

The change in the standard ESP allocation protocol to include the ES(D)P algorithm was proposed by the Eurotransplant Kidney Advisory Committee, based on the results of the 5-year analysis of the ESP.¹¹ The algorithm was set up to evaluate the need to reintroduce the previous standard of (at least minimal) matching for HLA antigens, the founding principle of kidney exchange within the Eurotransplant region. The decision was made by the international board, in close cooperation with the relevant regulatory authorities in the participating countries.

Only nonimmunized (≤5% panel-reactive antibodies) recipients waiting for a first kidney transplant and negative cellular cross-match before transplantation were included in the allocation algorithm. Participation did not interfere with the choice of treatment, sample collection, or medical procedures, which entirely followed standard hospital practice. The power calculation was based on the increased (treated) acute rejection rate observed in the 5-year ESP analysis.¹¹ To detect a 10% reduction with 80% power required allocation of at least 251 kidneys per arm. Numerous legal and logistic restrictions, however, made a trial design with rejection as the primary endpoint (including uniform clinical immunosuppression and central monitoring of biopsy-confirmed acute rejection) not feasible. We therefore concentrated on the mandatory allocation and survival data as captured by a dedicated coworker within the ESP registry. These reports do not include standard or certified data on acute rejection and estimated glomerular filtration rate. In parallel, the ESP program continued allocation of single DBD kidneys from elderly donors or donation-after-circulatory-death (DCD) donors to ESP candidates, including those with panel-reactive antibodies and/or repeat transplants.

Data source and ethical approval

Donor characteristics: age, sex, history of hypertension or diabetes, cause of death, intensive care unit stay, serum creatinine level, procurement procedure, HLA-typing, cold ischemia time, and recipient information (age, sex, cause of kidney failure, date of first dialysis, HLA-typing, panel reactive antibody status) were retrieved from the Eurotransplant Information System. Participation in the registry is mandatory, and next to baseline donor and recipient characteristics, follow-up data are collected annually for every kidney transplant (date of kidney graft failure and return to dialysis or patient death). During the paired kidney allocation period, a dedicated study coordinator collected and verified the data on patient and graft survival by means of phone calls once every 3 months to the participating transplant centers in 4 countries. The predefined follow-up period was 5 years. Patients were censored in cases in which loss to follow-up or end of follow-up (December 16, 2019) occurred. After the most recent annual ESP update-December 14, 2022-<2% of transplant recipients were lost to follow-up at 5 years. The study was approved by the medical ethical committee of the Leiden University Medical Center and the relevant authorities and organ procurement organizations in the participating ESP countries (Austria, Belgium, Germany, Netherlands). Patients provided written consent for the participation in the ESP/ESDP program.

Outcome parameters

Survival of the patient was defined as the time from transplantation until death. Mortality (with or without graft function) was ascertained at the transplant-center level by annual survey. Kidney graft failure was defined by return to dialysis (or a repeat transplantation). Primary nonfunction was defined as failure of the graft to ever function, irrespective of cause. Dialysis vintage was defined by the waiting time while on dialysis until transplantation, starting at the first day of dialysis initiation. The kidney donor–risk index (KDRI; with the following parameters: race, donor age, weight, height, hepatitis C virus antibodies, hypertension, diabetes, deceased braindeath donor) was used as a surrogate for quality of the deceaseddonor organ.²¹

Statistical analysis

Univariable comparisons between the ESDP and ESP group were done using χ^2 tests for categorical data, and *t* tests or Wilcoxon tests for parametric continuous and nonparametric continuous data, respectively. We used the Kaplan–Meier method to calculate the overall mortality incidence. We assessed the cause-specific risk of kidney graft failure using the competitive risk-analysis method (R statistics "survival" software package version 1.4.2; R Foundation). The *P* values were determined by log-rank test. Cox proportional hazards regression with list-wise deletion and forward selection (entry *P* = 0.05; removal *P* = 0.10), according to the standard SPSS (IBM) algorithm, was used to add covariables for multivariable analyses. We considered 2-sided *P* values <0.05 to be statistically significant. These analyses were performed using SPSS 25.0 (IBM Corp.).

RESULTS

Allocation, donor, and recipient characteristics

A total of 675 kidney transplant procedures were performed during the period of paired allocation within ESP; 341 kidneys were allocated with a zero HLA-DR mismatch (ESDP), and 334 kidneys were allocated according to waiting time, without HLA-DR matching, to ESP recipients (Supplementary Figure S1A). Five-year follow-up data were missing for 8 ESP recipients (2.4%) and 5 ESDP recipients (1.5%), respectively (Supplementary Figure S1B). In total, 662 patients were included in the complete case analysis. Almost 90% of kidneys were allocated in pairs. From 38 donors, only one kidney was transplanted, 22 were allocated via ESP, and 18 were allocated via ESDP. From 12 kidney donors, both kidneys were allocated to a DR-compatible recipient, and in 7 donors, both kidneys were allocated to ESP. Demographic and clinical characteristics of kidney donors and recipients are summarized in Table 1. Per the allocation algorithm, donor characteristics were paired, and all ESDP patients received a zero HLA-DR mismatched kidney. Likewise, the waiting time or median dialysis vintage (4.1 vs. 2.6 years) was significantly longer in the ESP group (P < 0.0001). The allocation strategy with HLA-DR matching resulted in a longer median cold ischemic time (12.0 hours [interquartile range {IQR}: 9.4–15.1) versus

Table 1 | Allocation of paired kidneys within ESP by matching for HLA-DR antigens

Characteristic	ESDP ($n = 336$)	ESP (<i>n</i> = 326)	Р
Allocation			
HLA-mismatch			
DR 0/1/2	100/0/0 ^a	0/49/51	
B 0/1/2	5/44/51	1/32/67	
A 0/1/2	13/55/32	12/44/44	
Waiting time/dialysis	2.6 (1.8–4.0) ^a	4.2 (2.8–5.5)	
vintage, yr			
Categorical, yr			
<2	30	13	
2–5	58	54	
>5	12	33	
Cold ischemia, h	12 (9–15) ^a	11 (8–13)	
Donor			
Age, yr	71 (68–74)	71 (68–74)	0.58
Sex, female	57	56	0.88
Cause of death			0.93
CVA	46	45	
SAB	22	25	
Trauma	12	13	
Anoxia	7	6	
Brain tumor	3	4	
Other/unknown	10	7	
Hospital stay, d	5 (2–7)	5 (2–7)	0.87
Creatinine, μmol/l	84 ± 45	85 ± 49	0.69
Kidney, left	50	50	0.94
KDRI score			0.95
1.00–1.49	19	18	
1.50–1.99	55	56	
≥2.00	26	26	
Recipient			
Age, yr	69 (67–72)	69 (66–72)	0.24
Sex, male	71	72	0.67
Native kidney disease			0.98
Glomerular/immunologic	27	31	
Diabetes/hypertension/ vascular	25	25	
Congenital/hereditary	13	12	
Tubulointerstitial/	6	5	
pyelonephritis			
Other	12	11	
Unknown	17	17	
Preemptive transplant	3	1	0.11

CVA, cerebrovascular accident; ESP, Eurotransplant Senior Program; ESDP, Eurotransplant Senior DR-compatible Program; HLA, human leukocyte antigen; KDRI, kidney donor risk index; SAB, subarachnoidal bleeding. $^{a}P < 0.0001$.

Values are %, median (interquartile range), or mean ($\pm \text{SD}$), unless otherwise indicated.

10.6 hours (IQR: 7.9–13.4) in ESDP and ESP (P < 0.0001), respectively. Of note, 50% of patients in the ESP arm of the allocation scheme received a kidney with 1 HLA-DR mismatch. Allocation within the ESP program is ABO-identical by default (ABO-O 54%; ABO-A 39%; ABO-B 6%; ABO-AB 1%). For this temporary change with paired allocation within the ESP program, only nonimmunized candidates panel reactive antibodies (PRA) <5% listed for a first kidney transplant were considered. Recipient and transplant characteristics were similar in the 2 groups.

Although female patients appeared to be somewhat underrepresented, the differences were minor, compared to all ESP candidates. Before initiation of the paired allocation within ESP (December 31, 2009), a total of 2131 candidates were listed in ESP. Among the 50% of candidates on the active waiting list, 88% were nonimmunized, with 67% being male and 33% female. Near the end (December 31, 2013), the corresponding numbers were 84%, 66%, and 34%, respectively. Although the percentage of immunized patients on the active waiting list increased between December 2009 and December 2013, from 12.2% to 16.4%, the proportion of female patients decreased from 60% to 53%. The incidences of primary nonfunction (4.9% vs. 6.4%) and delayed function were comparable in the ESDP and ESP recipients. Detailed information on the incidence of acute rejection and kidney function was not available. Initial immunosuppression showed the use of induction in 64% of patients (basiliximab in 93% of cases), and triple maintenance therapy consisting of steroids, calcineurin inhibitors (tacrolimus 69%), and mycophenolate (99%).

Survival analysis

Allocation of paired kidneys with a zero HLA-DR mismatch was associated with a significantly lower chance of mortality 5 years after transplantation (log-rank P < 0.005). The 5-year Kaplan–Meier estimates for (all-cause) mortality are plotted in Figure 1a. Besides a zero HLA-DR mismatched kidney transplant (hazard ratio [HR] 0.71; 95% confidence interval [CI] 0.53–0.95, P < 0.05), recipient age, end-stage (native) kidney failure due to arterial hypertension or diabetes, and return to dialysis in the first year post-transplant were independently associated with 5-year mortality (Table 2). The impact of kidney graft failure and return to dialysis in the first year after transplantation on 5-year mortality in this cohort of elderly transplant recipients is illustrated in Supplementary Figure S2.

The 5-year incidences of kidney graft failure are plotted in Figure 1b. The cause-specific hazards for kidney graft failure for allocation with a zero HLA-DR mismatched at 1 year and 5 years post-transplant were 0.55 (95% CI 0.35–0.87; P <0.01) and 0.73 (95% CI 0.53–0.99; P < 0.05), respectively (Tables 3 and 4). At 1-year allocation with a zero HLA-DR mismatch, donor as well as recipient age was associated with kidney graft failure. Multivariable analysis revealed the KDRI to be a parameter that was associated with 5-year kidney graft failure. The paired kidney allocation algorithm selected only DBD donors, first transplants, and no hepatitis C virus-positive candidates' antibodies were transplanted. The majority (approximately 75%) of patients received a kidney with a KDRI score less than 2 (Table 1). In this subgroup, the beneficial impact of a zero HLA-DR mismatched kidney was evident (log-rank P < 0.05; Figure 2) and was most pronounced in the first year after transplantation (logrank P = 0.015). Beyond the first year, the adverse impact of a higher KDRI score remained, but a beneficial effect of the allocation with a zero HLA-DR mismatch was no longer observed (Table 4).



Figure 1 | (a) Five-year (all-cause) mortality (Kaplan-Meier estimates, P < 0.01, hazard ratio [HR] 0.68 (95% confidence interval [CI] 0.51–0.91) and (b) kidney graft failure with cause-specific hazards at 1-year: HR 0.55 (95% CI 0.35–0.87); P < 0.01 and at 5 years: HR 0.71 (95% CI 0.52–0.98); P < 0.05, according to allocation of paired kidneys with (Eurotransplant Senior DR-compatible Program [ESDP]) or without (Eurotransplant Senior Program [ESP]) matching for human leukocyte antigen (HLA)–DR antigens. Parentheticals give 95% Cl.

Waiting time/dialysis vintage

Allocation in ESP is primarily directed by waiting time while on dialysis. Dialysis vintage was 4.1 years (IQR 2.81–5.49) with the ESP scenario, versus 2.6 years (IQR 1.85–3.96) via ESDP (P < 0.00005; Table 1). The proportion of patients who received a transplant within 5 years after initiation of dialysis was 88% in the ESDP group, versus less than 67% for ESP (Table 1). The ESDP scenario resulted in significantly more transplants in elderly patients who spend less than 2 years on dialysis (30% vs. 13% in ESP; P < 0.0001). The cumulative incidences in kidneys offered and transplanted, by waiting time, are plotted in Figure 3a. Median waiting times were

different in Austria, Belgium, Germany, and Netherlands, but the cumulative incidences of elderly kidney transplants by dialysis vintage remained comparable over the years of allocation (Figure 3b). By principle, the ESP selected patients based on longest waiting time, resulting in 33% of kidneys being allocated to elderly recipients with a dialysis vintage over 5 years versus only 13% in the ESDP scenario (zero HLA-DR mismatched, followed by the longest waiting time). Cumulative incidences of mortality by dialysis vintage, according to allocation with or without HLA-DR mismatch, are plotted in Figure 3. No significant differences were found for waiting times up to 5 years. Beyond a 5-year dialysis vintage,

Table 2 | Five-year mortality

Factor	Hazard ratio (95% CI)			
	Univariable	Р	Multivariable	Р
Allocation				
Allocation via ESDP vs. ESP	0.69 (0.52–0.92)	<0.05	0.71 (0.53–0.95)	<0.05
Dialysis vintage, yr	1.09 (1.01–1.17)	<0.05	1.07 (1.01–1.17)	< 0.05
HLA-class-I mismatch	1.03 (0.89–1.21)	0.68		
Donor age, yr	1.01 (0.98–1.04)	0.77		
Donor sex, M/F	0.88 (0.66-1.18)	0.39		
Cold Ischemia, h	1.00 (0.97–1.03)	0.96		
Recipient				
Age, yr	1.04 (0.99–1.09)	<0.01	1.07 (1.02–1.11)	<0.01
Sex, M/F	1.02 (0.73–1.43)	0.89		
ESRD: hypertension or diabetes	1.40 (1.00–1.94)	<0.05	1.49 (1.07–2.09)	< 0.05
Return to dialysis in first year	3.46 (2.38–5.02)	<0.0001	3.33 (2.28–4.86)	< 0.0001

CI, confidence interval; ESDP, Eurotransplant Senior DR-compatible Program; ESP, Eurotransplant Senior Program; ESRD, end-stage renal disease; F, female; HLA, human leukocyte antigen; M, male.

Factors	Hazard ratio (95% CI)			
	Univariable	Р	Multivariable	Р
Allocation				
Allocation via ESDP vs. ESP	0.55 (0.35–0.87)	< 0.05	0.59 (0.38–0.93)	<0.05
Dialysis vintage, yr	1.13 (1.02–1.25)	<0.05	1.08 (0.97–1.21)	0.15
Donor age, yr	1.06 (1.02–1.11)	<0.01	1.06 (1.02–1.11)	< 0.01
KDRI score				
1.00–1.49	1		1	
1.50–1.99	2.00 (0.94-4.23)	0.07	1.81 (0.83-3.94)	0.13
≥2.00	2.52 (1.14–5.54)	< 0.05	1.84 (0.71–4.73)	0.21
Donor sex, M/F	0.89 (0.57-1.38)	0.59		
HLA-class-I mismatch	1.19 (0.93–1.52)	0.17		
Cold ischemia, h	0.98 (0.93–1.03)	0.41		
Recipient				
Age, yr	0.92 (0.86–0.98)	<0.05	0.92 (0.86–0.99)	<0.05
Sex, M/F	1.00 (0.71–1.43)	0.98		
Kidney failure (hypertension or diabetes vs. other causes): AHT or DM	1.26 (0.78–2.04)	0.35		

Table 3 | One-year kidney graft failure (death-censored)

AHT, arterial hypertension; DM, diabetes mellitus; CI, confidence interval; ESDP, Eurotransplant Senior DR-compatible Program; ESP, Eurotransplant Senior Program; F, female; HLA, human leukocyte antigen; KDRI, kidney donor risk index; M, male.

the cumulative incidences of kidneys offered and transplanted were similar (Supplementary Figure S3). Allocation with a zero HLA-DR mismatch according to ESDP, however, was associated with a significantly lower chance of mortality (Figure 3). No differences were present in candidate age, cause of their kidney failure, or in the donor age or KDRI score of the kidneys offered. Return to dialysis within the first postoperative year occurred significantly less frequently in those who received a kidney transplant w/o HLA-DR mismatches. The beneficial effect of a zero mismatched kidney on kidney graft failure in the context of the KDRI score is plotted in Figure 4.

DISCUSSION

The introduction of the ESP allocation program has significantly improved the chances that elderly waitlisted dialysis patients will receive a timely transplant. Although the majority of patients entering renal replacement programs are over the age of 65 years, or even 75 years, elderly patients who are placed on the waiting list represent a highly selected subgroup of transplant candidates.²³ In this paired kidney allocation study, we investigated for the first time the effect of HLA-DR matching on outcome within the ESP. The inclusion of prospective matching for HLA-DR antigens in the allocation algorithm was associated with a 30 percent lower chance

Table 4 | Kidney graft failure (censored): 5 years

Factors	Hazard ratio (95% CI)				
	Univariable	Р	Multivariable	Р	
Allocation					
Allocation via ESDP vs. ESP	0.71 (0.52–0.98)	< 0.05	0.73 (0.53–0.99)	< 0.05	
Dialysis vintage, yr	1.03 (0.96–1.13)		1.06 (0.98–1.15)	0.16	
Donor age, yr	1.06 (1.03–1.09)	< 0.0005	1.03 (0.99–1.08)	0.16	
KDRI score					
1.00–1.49	1		1		
1.50–1.99	2.04 (1.18–3.54)	<0.05	1.85 (1.05–3.26)	< 0.05	
≥2.00	2.88 (1.62-5.12)	< 0.0005	2.17 (1.08-4.36)	< 0.05	
Donor sex, M/F	0.82 (0.59–1.14)	0.24			
HLA-class-I mismatch	1.03 (0.87-1.23)	0.73			
Cold ischemia, h	1.01 (0.98–1.04)	0.54			
Recipient					
Age, yr	0.97 (0.92–1.02)	0.20			
Sex, M/F	1.00 (0.71–1.43)	0.98			
Kidney failure (hypertension or diabetes vs. other causes): hypertension/diabetes	1.24 (0.87–1.78)	0.24			

CI, confidence interval; ESDP, Eurotransplant Senior DR-compatible Program; ESP, Eurotransplant Senior Program; F, female; HLA, human leukocyte antigen; KDRI, kidney donor risk index; M, male.



Figure 2 | Cumulative incidence of transplanted candidates by dialysis vintage according to (a) allocation of paired kidneys with (Eurotransplant Senior DR-compatible Program [ESDP]) or without (Eurotransplant Senior Program [ESP]) matching for human leukocyte antigen (HLA)-DR antigens and (b) by calendar year. IQR, interquartile range.

of mortality 5 years after transplantation and more functioning kidney transplants. In particular, 1-year deathcensored kidney graft survival was significantly better and could eventually explain a lower mortality incidence, as return to dialysis in the first post-transplant year was an important determinant for patient death after 5 years. The allocation algorithm based on HLA-DR matching resulted in a shift in waiting time, with significantly more recipients receiving a kidney offer within the first 2 years after initiation of dialysis. Of importance, the benefit of prospective HLA-DR matching was independent of other factors that were associated with all-cause mortality, including dialysis vintage.



Figure 3 | Five-year (all-cause) mortality by dialysis vintage (<2 year; 2–5 years; ≥5 years) according to allocation of paired kidneys with (Eurotransplant Senior DR-compatible Program [ESDP]) or without (Eurotransplant Senior Program [ESP]) matching for human leukocyte antigen (HLA)–DR antigens.



Figure 4 Kidney graft failure by initial quality assessment (a) kidney donor risk index (KDRI) score <2 and (b) KDRI score ≥ 2 according to allocation of paired kidneys with (Eurotransplant Senior DR-compatible Program [ESDP]) or without (Eurotransplant Senior Program [ESP]) matching for human leukocyte antigen [HLA]-DR antigens.

Return to dialysis in the first post-transplant year was strongly associated with excess 5-year mortality, with comparable hazards in recipients of kidneys allocated with versus without HLA-DR matching. Kidney allograft failure in the context of old-for-old allocation may have immunologic and nonimmunologic determinants or both. First, it has become widely accepted that recipients of older kidneys experience higher acute rejection rates within each recipient age category.^{15–17,20} Episodes of T cell–mediated acute rejection are associated with histo-incompatibility, and in particular, the degree of mismatching for HLA-DR antigens.^{18,19}

Although we were unable to retrieve detailed information on acute rejection from the registry, the large majority of acute rejection episodes in this (nonimmunized) cohort of older first kidney–transplant recipients of low to standard immunologic risk were most likely Tcell–mediated. These episodes generally respond well to standard rejection treatment strategies, but functional consequences or reversibility may also depend on the tissue quality of the kidney transplanted. In this cohort of kidney donors with a median donor age of 72 years, histocompatibility for HLA- DR antigens resulted in significantly better 1-year kidney allograft survival. In particular, a more pronounced benefit was found in recipients of kidneys with KDRI scores below 2. The large majority of the patients received a kidney graft with a KDRI score of less than 2. Thus, these results support the principal benefit of preventing acute rejection by histocompatibility, and the need of additional courses of high-dose corticosteroids and/or T cell-depleting antibodies, which result in excess infection-related morbidity and mortality, especially in the elderly. Acute rejection may be adequately treated, but functional reversibility in vulnerable kidneys is likely to be incomplete, while increasing the risk of infection-related mortality. Indeed, although the 5-year analysis of the ESP data has documented quite acceptable kidney graft survival rates, the rejection rate was 5%-10% higher, and mortality due to infections was 30%, which is quite high.¹¹

By default, the ESP program candidates are selected according to their having an identical ABO blood group and their cumulative waiting time (i.e., dialysis vintage) starting the day of first dialysis. Consequently, a significant proportion of kidneys in the different countries were offered to candidates with a dialysis vintage of over 5 years. The uniform finding in recent literature has been that dialysis vintage has a consistently impact on patient survival and uncensored graft survival.^{10,24} In addition, with time, candidates may experience more frequent (temporary) delisting (or may die) before a kidney becomes available. The use of HLA-DR compatibility as the principal selection parameter resulted in a significant shift in the median dialysis vintage, and in significantly more kidneys being offered to recipients with a dialysis vintage of less than 2 years (Table 1). More than one explanation for this observation may be possible, but the most plausible in our opinion is as follows. The chance of a receiving a kidney offer is most likely a direct reflection of the relative ABO blood group and the HLA class-II allele or haplotype frequencies in the background population of both recipients and potential kidney donors. Finding a zero mismatch between donor and recipient is facilitated by significantly less polygenic and polymorphic characteristics of HLA-DR, as compared to HLA class-I molecules. Such a "reset" by the allocation policy is supported by at least the observation that differences in waiting time with or without prospective matching for HLA-DR antigens occurred within the first year of randomized allocation and structurally overruled waiting time in the following years. A better match with an earlier transplant reduces the chance of being (temporarily) delisted for those candidates and of not receiving a kidney offer. Given that transplant numbers in this selected pool of elderly candidates are not increasing, the impact of delisting is likely to remain constant, but it may include factors such as sex and/or sensitization. Overall, we found similar delisting parameters for male patients and female patients, and/or immunization. A detailed evaluation of (temporary) delisting that includes all male and female ESP patients, as well as transplant candidates of minority ethnic groups, is beyond the scope of this article.

The current study also has limitations. In this analysis of registry data, we did not have complete and detailed information regarding the occurrence of acute rejection episodes, rejection treatment(s), or evolution of renal allograft function. Second, we cannot determine from this study whether female patients who are immunized or transplant candidates of minority ethnic groups will be disadvantaged by this strategy. In parallel, the ESP program continued to allocate single kidneys from elderly donors (DBD or DCD) to ESP candidates, immunized or not, and/or repeat transplant candidates.

In conclusion, this paired kidney allocation algorithm within the ESP documented an independent beneficial effect of HLA-DR matching on the incidences of both mortality and functioning kidney grafts in the context of old-for-old kidney transplantation. In addition, kidney allocation based on HLA-DR matching resulted in a small increase in cold ischemia and shorter dialysis duration.

APPENDIX

Members of the Eurotransplant Senior DR-compatible Program (ESDP) Study Group

I. Tieken, G. Haasnoot, M. van Meel Eurotransplant (Registry and Reference Lab); L.C. Rump, Düsseldorf; A. Rosenkranz, S. Horn, Graz; R. Margreiter, S. Schneeberger, Innsbruck; R. Oberbauer, E. Pohanka, F. Függer, Linz; F. Mühlbacher, G. Berlakovich, Vienna; M. Meurisse, L. Weekers, Liége; D. Ysebaert, Antwerp; K.M. Wissing, D. Mikhalski, M. Mourad, Brussel; W. van Biesen, Gent; D. Kuypers, Leuven; J. Floege, Aachen; M. Anthuber, Augsburg; R. Viebahn, P. Schenker, Bochum; K. Budde, J. Pratschke, W. Zidek, Berlin; S. Melchior, Bremen; R. Woitas, C.H. Strassburg, Bonn; C. Hugo, M. Wirth, Dresden; M. Schiffer, Nurnberg; A. Kribben, Essen; P. Pisarski, S. Fichtner-Feigl, Freiburg; M. Haubitz, Fulda; R. Weimer, Giessen; P. Weithofer, Gottingen; P. Fornara, Halle; L. Fisher, Hamburg; U. Sester, Homburg; M. Zeier, Bad Nauheim; V. Kliem, I Klempnauer, Hannover; M.O. Grimm, Jena; U. Kunzendorf, Kiel; D. Stippel, W. Arns, C. Mönch, Cologne; M. Nitschke, Lubeck; M. Bartels, Leipzig; B. Krämer, B. Kruger, Mannheim; U. Heemann, J. Werner, München; J. Hoyer, Marburg; H.H. Wolters, B. Suwelack, Münster; J. Lutz, Mainz; B. Banas, Regensburg; O. Hakenberg, Rostock; C.J. Olbricht, M. Kalus, V. Schwenger, Stuttgart; S. Nadalin, Tubingen; B. Schröppel, Ulm; K. Lopau, Wurzburg; M.A.J. Seelen, S. Berger, Groningen; J. de Fijter, Leiden; S.J. van der Linden, M.H.L. Christiaans, Maastricht; J. van de Wetering, Rotterdam; A.D. van Zuilen, Utrecht; F. Bemelman, A. Nurmohamed, Amsterdam; and L. Hilbrands, Niimegen.

DISCLOSURE

All the authors declared no competing interests.

SUPPLEMENTARY MATERIAL

Supplementary File (PDF)

Supplementary Figure S1. (**A**) Flow chart of the paired allocation algorithm within the Eurotransplant Senior Program. (**B**) Transplant recipients included in the follow-up analyses.

Supplementary Figure S2. Cumulative incidence of (all-cause) mortality in "old-for-old" kidney transplant recipients, stratified by death with functioning kidney graft or return to dialysis within the first post-transplant year.

Supplementary Figure S3. Kidneys transplanted by dialysis vintage <5 or \geq 5 years according to allocation of paired kidneys with (Eurotransplant Senior DR-compatible Program [ESDP]) or without (Eurotransplant Senior Program [ESP]) matching for human leukocyte antigen (HLA)–DR antigens.

REFERENCES

- 1. European Renal Association–European Dialysis and Transplant Association (ERA-EDTA) Registry. *ERA-EDTA registry annual report 2017*. Amsterdam UMC; 2019.
- United States Renal Data System. 2019 USRDS annual data report: Epidemiology of kidney disease in the United States. National Institutes of Health; 2019.
- 3. Lenain R, Boucquemont J, Leffondré K, et al. Clinical trial emulation by matching time-dependent propensity scores: the example of estimating impact of kidney transplantation. *Epidemiology*. 2021;32:220–229.
- Meier-Kriesche HU, Ojo AO, Port FK, et al. Survival improvement among patients with end-stage renal disease: trends over time for transplant recipients and wait-listed patients. J Am Soc Nephrol. 2001;12:1293–1296.
- 5. Jofre R, Lopez-Gomez JM, Moreno F, et al. Changes in quality of life after renal transplantation. *Am J Kidney Dis.* 1998;32:93–100.
- 6. Jarl J, Desatnik P, Peetz Hansson U, et al. Do kidney transplantations save money? A study using a before-after design and multiple register-based data from Sweden. *Clin Kidney J.* 2018;11:283–288.
- Senanayake S, Graves N, Healy H, et al. Donor kidney quality and transplant outcome: an economic evaluation of contemporary practice. *Value Health.* 2020;23:1561–1569.
- Rao PS, Merion RM, Ashby VB, et al. Renal transplantation in elderly patients older than 70 years of age: results from the Scientific Registry of Transplant Recipients. *Transplantation*. 2007;83:1069–1074.
- 9. Haller MC, Kammer M, Oberbauer R. Dialysis vintage and outcomes in renal transplantation. *Nephrol Dial Transplant*. 2019;34:555–560.

- 10. Prezelin-Reydit M, Combe C, Harambat J, et al. Prolonged dialysis duration is associated with graft failure and mortality after kidney transplantation: results from the French transplant database. *Nephrol Dial Transplant*. 2019;34:538–545.
- Frei U, Noeldeke J, Machold-Fabrizii V, et al. Prospective age-matching in elderly kidney transplant recipients—a 5-year analysis of the Eurotransplant Senior Program. Am J Transplant. 2008;8:50–57.
- 12. Kauffman HM, McBride MA, Cors CS, et al. Early mortality rates in older kidney recipients with comorbid risk factors. *Transplantation*. 2007;83: 404–410.
- **13.** Lemoine M, Titeca Beauport D, Lobbedez T, et al. Risk factors for early graft failure and death after kidney transplantation in recipients older than 70 years. *Kidney Int Rep.* 2019;4:656–666.
- Meier-Kriesche HU, Ojo AO, Hanson JA, Kaplan B. Exponentially increased risk of infectious death in older renal transplant recipients. *Kidney Int.* 2001;59:1539–1543.
- de Fijter JW, Mallat MJ, Doxiadis II, et al. Increased immunogenicity and cause of graft loss of old donor kidneys. J Am Soc Nephrol. 2001;12: 1538–1546.
- **16.** Oberhuber R, Ge X, Tullius SG. Donor age-specific injury and immune responses. *Am J Transplant*. 2012;12:38–42.
- 17. Tullius SG, Tran H, Guleria I, et al. The combination of donor and recipient age is critical in determining host immunoresponsiveness and renal transplant outcome. *Ann Surg.* 2010;252:662–674.

- Halleck F, Khadzhynov D, Liefeldt L, et al. Immunologic outcome in elderly kidney transplant recipients: Is it time for HLA-DR matching? *Nephrol Dial, Transplant.* 2016;31:2143–2149.
- Halloran PF, Homik J, Goes N, et al. The "injury response": a concept linking nonspecific injury, acute rejection, and long-term transplant outcomes. *Transplant Proc.* 1997;29:79–81.
- 20. Peters-Sengers H, Berger SP, Heemskerk MB, et al. Stretching the limits of renal transplantation in elderly recipients of grafts from elderly deceased donors. *J Am Soc Nephrol.* 2017;28:621–631.
- 21. Rao PS, Schaubel DE, Guidinger MK, et al. A comprehensive risk quantification score for deceased donor kidneys: the kidney donor risk index. *Transplantation*. 2009;88:231–236.
- 22. Vogelzang JL, van Stralen KJ, Noordzij M, et al. Mortality from infections and malignancies in patients treated with renal replacement therapy: data from the ERA-EDTA registry. *Nephrol Dial Transplant*. 2015;30: 1028–1037.
- 23. Beuscart JB, Pagniez D, Boulanger E, Duhamel A. Registration on the renal transplantation waiting list and mortality on dialysis: an analysis of the French REIN registry using a multi-state model. *J Epidemiol*. 2015;25: 133–141.
- 24. Legeai C, Andrianasolo RM, Moranne O, et al. Benefits of kidney transplantation for a national cohort of patients aged 70 years and older starting renal replacement therapy. *Am J Transplant*. 2018;18: 2695–2707.